Author's response to reviews

Title: Germinated brown rice and its bioactives modulate the activity of uterine cells in oophorectomised rats as evidenced by gross cytohistological and immunohistochemical changes

Authors:

Sani M Ismaila (snismaila99@gmail.com)
Maznah Ismail (maznah@medic.upm.edu.my)
Rozi B Mahmud (rozi@medic.upm.edu.my)
Abubakar M Salisu (msabuus@yahoo.com)
MD Zuki Abu Bakar Zakaria (mdzukiabubakar@gmail.com)

Version: 3 Date: 24 June 2013

Author's response to reviews: see over
RESPONSE TO REVIEWERS COMMENTS

REVIEWER 1

1. Punctuation errors were corrected
2. The article has been proof read by a professional proof reader

REVIEWER 2

1. Contrast and resolution of image 3 have been improved
2. Methods:
   - Line 3: dehydrated in graded alcohol was changed to rehydrated in graded alcohol
   - Line5: “unspecific bindings were covered using pen was changed to; unspecific bindings were covered using 5% Bovine serum albumin (5% BSA)
   - anti body was changed to antibody
   - “data are presented” was changed to “data were presented